Author's response to reviews

Title: Elevated levels of circulating microRNA-200 family members correlate with serous epithelial ovarian cancer

Authors:

Casina WS Kan (casina.kan@sydney.edu.au)
Michael A Hahn (m.hahn@griffith.edu.au)
Gregory B Gard (ggard@med.usyd.edu.au)
Jayne Maidens (JMaidens@nsccahs.health.nsw.gov.au)
Jung Yoon Huh (jyhuh07@gmail.com)
Deborah J Marsh (deborah.marsh@sydney.edu.au)
Viive M Howell (viive.howell@sydney.edu.au)

Version: 2 Date: 29 September 2012

Author's response to reviews: see over